Author's response to reviews

Title: Dual method contraception among Human Immunodeficiency Virus positive parturient in south eastern Nigeria

Authors:

LUCKY O LAWANI (lawkins2000@yahoo.com)
CHUKWUEMEKA A IYOKE (caiyoke@yahoo.uk.co)
ONYEBUCHI A KANARIO (kanayo009@yahoo.com)

Version: 2 Date: 28 October 2013

Author's response to reviews: see over
Cover letter

Dear editor in chief,

Please, find enclosed our manuscript titled “Dual method contraception among HIV positive parturient in southeastern Nigeria” by Lawani OL, Iyoke CA and Onyebuchi AK which we are re-submitting for publication as an article in BMC Women’s Health. The tables have been included in the manuscript after the reference section as recommended.

Thank you.

We look forward to hearing from you at your earliest convenience.

Kind regards.

Dr. Lawani OL